Author's response to reviews

Title: Risk factors for eczema in infants born in Cuba: a population-based cross-sectional study

Authors:

Ramon Suarez-Medina (ramonsm@inhem.sld.cu)
Silvia Venero-Fernandez (silviaf@infomed.sld.cu)
Esperanza Mora-Faife (mora@infomed.sld.cu)
Ileana Valle-Infante (idelvalle@infomed.sld.cu)
Gladys Garcia-Garcia (gladysgarcia@infomed.sld.cu)
Liem Gomez-Marreo (liemgomez@infomed.sld.cu)
Dania Fabre-Ortiz (fabre@infomed.sld.cu)
Hermes Fundora-Hernandez (hermes.fundora@infomed.sld.cu)
Andrea Venn (andrea.venn@nottingham.ac.uk)
John Britton (j.britton@virgin.net)
Andrew Fogarty (andrew.fogarty@nottingham.ac.uk)

Version: 5
Date: 4 March 2014

Author's response to reviews:

Dear Ms Partridge,

I have adjusted the Cuban authors’ initials as requested. The problem was a consequence of the revision of the manuscript to allow for the Spanish style family names.

I hope that this solves the problem and please contact me if there are any further queries.

Yours sincerely,
Andrew